Author's response to reviews

Title: Colectomy rate in steroid-refractory colitis initially responsive to cyclosporin: A long-term retrospective cohort study

Authors:

Giovanni C Actis (actis_g@libero.it)
Maurizio Fadda (mfadda@molinette.piemonte.it)
Ezio David (edavid@molinette.piemonte.it)
Anna Sapino (asapino@molinette.piemonte.it)

Version: 5 Date: 21 March 2007

Author’s response to reviews: see over
To the kind attention of The BioMed Central Editorial Production Team:

According to your requirements, we are herein including the formatted copy of our manuscript.

1) The affiliations were revised
2) The presentation details were removed from the front page and incorporated in the acknowledgments
3) The heading Abstract was added
4) “Methods” in the Abstract
5) Key words section removed
6) “Methods” in the manuscript text
7) “Abbreviations” simply
8) Sentence added on Authors contributions
9) “Figure legends”
10) “Figures” modified

Hoping these changes fit your requirements, we send you the

Best regards

GC Actis, MD